Author’s response to reviews

Title: Air pollution exposure is associated with MRSA acquisition in young U.S. children with cystic fibrosis

Authors:

Kevin Psoter (kpsoter1@jhu.edu)
Margaret Rosenfeld (margaret.rosenfeld@seattlechildrens.org)
Anneclaire De Roos (ajd335@drexel.edu)
Jon wakefield (jonno@uw.edu)
Jonathan Mayer (jmayer@u.washington.edu)

Version: 2 Date: 03 Jul 2017

Author’s response to reviews:

Thank you for your detailed review of this manuscript by the Editors and Reviewer. We have addressed the comment by the Associate Editor below.

In the abstract, page 2, the sentence "PM 2.5 was not associated with acquisition of other respiratory pathogens." should be moved from the "Conclusions" (line 46) to the "Results" (line 41).

In the Abstract, we have moved the sentence from the Conclusions to the Results section.